Author's response to reviews

Title: Prevalence of prehypertension and its relationship to risk factors for cardiovascular disease in Jamaica: analysis from a cross-sectional survey

Authors:

Trevor S Ferguson (trevor.ferguson02@uwimona.edu.jm)
Novie O Younger (novie.younger@uwimona.edu.jm)
Marshall K Tulloch-Reid (marshall.tullochreid@uwimona.edu.jm)
Marilyn B Lawrence Wright (marilyn.lawrencewright@uwimona.edu.jm)
Elizabeth M Ward (warde@moh.gov.jm)
Deanna E Ashley (deca@cwjamaica.com)
Rainford J Wilks (rainfordw@cwjamaica.com)

Version: 5 Date: 9 August 2008

Author's response to reviews:

Dear Editorial Production Team

We have made the major formatting changes requested. Among the "minor changes" requested we have done all except for of including all the authors when these are less than 30. As offered in the message we would appreciate if the editors can make this change for us.

Thank you for accepting the manuscript.

Yours truly

Rainford Wilks.